Contents lists available at ScienceDirect

Cities

journal homepage: www.elsevier.com/locate/cities

Ride-hailing and transit accessibility considering the trade-off between time and money

accessibility and equity studies.

Rafael H.M. Pereira^{a,*}, Daniel Herszenhut^a, Marcus Saraiva^a, Steve Farber^b

^a Institute for Applied Economic Research – Ipea, Brazil

^b University of Toronto, Canada

ARTICLE INFO ABSTRACT Keywords: Ride-hailing services can expand access to opportunities in urban areas, but out-of-pocket costs may limit its **Ride-hailing** benefits for low-income individuals. This paper examines how ride-hailing shapes spatial and socioeconomic Accessibility differences in access to opportunities while accounting for the trade-off between travel time and monetary costs. Equity Using one year of aggregate Uber trip data for Rio de Janeiro in 2019 and a new multi-objective routing algo-Rio de Janeiro rithm, we analyze the potential for ride-hailing services to improve employment accessibility when used as a First-mile transit standalone transportation mode and in conjunction with transit as a first-mile connection. We find that, Affordability compared to transit, standalone ride-hailing can significantly expand accessibility as a standalone mode for short Pareto frontier trips, and as a first-mile feeder to transit in trips longer than 30 min. However, the accessibility benefits of ridehailing accrue mostly to high-income groups due to affordability barriers. These findings suggest that policy efforts to integrate rideshare with transit are likely not going to benefit low-income communities without some form of subsidized fare discounts to alleviate affordability barriers. The paper also highlights how accounting for

1. Introduction

The emergence of ride-hailing services has been one of the most disruptive transportation innovations in recent decades (Chan & Shaheen, 2012; Dudley et al., 2017; Tirachini, 2020). Transportation network companies (TNCs), such as Uber and DiDi, can help people overcome transit network gaps and improve urban accessibility without the costs of car ownership (Brown et al., 2022; Jin et al., 2018; Young & Farber, 2019). Nonetheless, even though ride-hailing services are generally faster and more convenient than transit options, they are also more expensive for single riders and out-of-pocket costs are one of the biggest barriers to ride-hail for low-income people (Brown et al., 2022). This explains in part the growing number of studies trying to understand the circumstances under which ride-hailing could compete and complement public transit (Cats et al., 2022; Hall et al., 2018; Li et al., 2021; Yan et al., 2019; Young et al., 2020).

Various studies have shown how ride-hailing services can effectively expand mobility options (Alemi et al., 2018; Ceccato & Diana, 2021; Onono et al., 2019; Tirachini & del Río, 2019). However, few studies have examined the extent to which ride-hailing services improve access

to opportunities, particularly when used in conjunction with transit, and how such accessibility benefits can vary across socioeconomic groups. Moreover, no studies have compared ride-hailing and transit accessibility considering the trade-off between travel time and monetary costs. Previous research on accessibility by ride-hailing has focused on either travel time (Barkley et al., 2018; Haddad et al., 2019) or out-of-pocket costs (Souza et al., 2021). Two exceptions are the works of Abdelwahab et al. (2021) and Cats et al. (2022). Nonetheless, these studies use generalized travel cost functions that combine time and monetary costs into a single cost value, which ignores the potential trade-offs between travel time and monetary costs (see Section 2). Being able to examine these trade-offs is particularly important for those investigating the potential competition and complementarity between ride-hailing and public transit because of how these transport modes provide markedly different levels of accessibility, though at significantly different costs (Schwieterman, 2019).

trade-offs between travel-time and monetary costs can importantly influence the results of transportation

The aim of this paper is twofold. First, it examines how ride-hailing services can shape spatial and socioeconomic differences in access to opportunities while accounting for the trade-off between travel time and monetary costs. Second, the paper investigates the extent to which urban

* Corresponding author. *E-mail address:* rafael.pereira@ipea.gov.br (R.H.M. Pereira).

https://doi.org/10.1016/j.cities.2023.104663

Received 14 April 2023; Received in revised form 8 November 2023; Accepted 13 November 2023 0264-2751/© 2023 Elsevier Ltd. All rights reserved.







accessibility can be expanded by ride-hailing when these services are used either as a standalone transport mode or when used in conjunction with transit as a first-mile feeder to mass transit. The paper draws on material from one year of aggregate Uber trip data for Rio de Janeiro (Brazil) in 2019. Combining this data with multimodal transport routing, we calculate employment accessibility using cumulative opportunity measures that simultaneously account for both travel time and monetary costs. This is done using a new method recently developed by Conway and Stewart (2019) and implemented in the R package r5r (Pereira et al., 2021), which provides a multiobjective optimization routing method to account for cost constraints in public transit accessibility metrics. This work also combines custom travel matrices by Uber and transit designed to accommodate transfer times between these modes, which are used to calculate accessibility levels that result from the combined use of transit and Uber as a first-mile feeder.

This study advances the literature on transportation accessibility and equity by examining the trade-off between travel time and out-of-pocket monetary costs in accessibility. Our case study advances the state-of-thepractice in transportation equity analysis by showing how explicitly considering affordability of different modes affects transport accessibility and inequality estimates. Additionally, it expands upon previous research by analyzing the potential for ride-hailing services to improve accessibility when used in conjunction with transit as a first-mile feeder service. This is particularly relevant given the increasing number of transit agencies partnering with TNCs to improve accessibility for transportation-disadvantaged communities in low-density areas (Brown et al., 2021; Curtis et al., 2019; Zuniga-Garcia et al., 2022). Furthermore, this paper contributes to the growing empirical evidence on how ridehailing services may shape transportation equity and accessibility conditions, providing new evidence in the context of the Global South.

The notion of transportation equity with regards to transport accessibility can be framed both in terms of egalitarian or sufficientarian perspectives, which focus respectively on issues of relative inequalities in access to opportunities, and on whether people have enough (absolute) level of access to opportunities (Lucas et al., 2016; Pereira et al., 2017). In this paper, we focus particularly on inequality issues, and take the predominant stance in the literature that a transport system is considered more equitable if it contributes to reducing inequalities of opportunities by prioritizing the accessibility of disadvantaged groups (Pereira et al., 2017; Zhang & Zhao, 2021).

The remainder of the paper is organized as follows. Section 2 presents a literature review on ride-hailing services and urban accessibility. Section 3 presents the study area analyzed in the paper. The data and methods used to calculate travel times and monetary costs and to estimate employment accessibility are presented in Sections 4 and 5. Section 6 presents the results, and Section 7 presents the conclusion of the study with reflections for future research and policy recommendations.

2. Literature review

Ride-hailing services are a relatively recent innovation in urban mobility, but they have been capturing a lot of attention from researchers and policy makers (Shaheen & Cohen, 2019). Several studies have analyzed the implications ride-hailing services can have for urban mobility systems in terms of public transit ridership (Diab et al., 2020; Erhardt et al., 2022; Lavieri & Bhat, 2019; Rayle et al., 2016) and transport externalities such as road congestion (Diao et al., 2021; Erhardt et al., 2019; Tirachini & Gomez-Lobo, 2020), air pollution (Barnes et al., 2020; Sui et al., 2019; Yu et al., 2017) and traffic crashes (Anderson & Davis, 2021; Barreto et al., 2021; Barrios et al., 2019; Conner et al., 2021; Kirk et al., 2020).

There is also a growing literature that examines how transportation network companies (TNCs), such as Uber, DiDi and Cabify, could have ambiguous effects as a complement and substitute to public transit (Hall et al., 2018; Yan et al., 2019; Young et al., 2020). While ride-hailing services can compete as a more convenient though more expensive

transport alternative, these services can also increase the reach and flexibility of public transport by providing first mile-last mile connections (Cats et al., 2022; Shaheen & Chan, 2016; Wang et al., 2012). A few studies in this literature try to assess this potential complementarity and competition by looking at the overlap between the origins and destinations of trips conducted by ride-hailing versus transit (Jin et al., 2019; Kong et al., 2020; Liao, 2021). Other studies using travel surveys have shown that TNCs are increasingly used as first-last mile alternatives to integrate with transit (Bedoya-Maya et al., 2022; Brown et al., 2021; Sunitiyoso et al., 2022). Along these lines, there is also growing research on ride-hailing and transportation equity, looking at whether TNCs can effectively improve the mobility conditions of low-income and disadvantaged communities (Brown et al., 2022; Jiao & Wang, 2020; Jin et al., 2019) or whether ride-hailing services end up exacerbating the gap in the urban mobility and accessibility conditions of different socioeconomic groups (Abdelwahab et al., 2021; Barajas & Brown, 2021; Brown et al., 2021). There is also a growing literature looking at gender issues, and which examines the extent to which ride-hailing services benefit women's access given gendered particularities in travel behavior and safety concerns (Qiao et al., 2023; Sabogal-Cardona et al., 2021; Young & Farber, 2021).

Previous research has analyzed the wait times of passengers from TNCs as a surrogate to access to ride-hailing services (Hughes & MacKenzie, 2016; Insardi et al., 2019; Shokoohyar et al., 2020; Wang & Mu, 2018; Young & Farber, 2020). However, these studies provide only a limited account of accessibility, because they overlook travel time and land use patterns. By doing so, these studies focus on access to ride-hailing itself, and not on the level of access to opportunities that can be achieved by using these services.

Only a few studies have examined the extent to which ride-hailing can improve people's access to opportunities. One of these studies is the work conducted by Souza et al. (2021), who analyze four districts of Rio de Janeiro (Brazil). The authors compare the number of jobs that can be accessed by public transport and Uber given different monetary cost limits and number of passengers traveling together. The authors find that, overall, ride-hailing services provide substantially higher employment accessibility than transit, but they only become financially competitive once trip costs are shared between two or more passengers. Despite the contributions of the paper in analyzing transportation equity with ride-hailing, this study ignores how travel times are substantially different in the shaping of accessibility by ride-hailing and transit.

The work of Cats et al. (2022) analyzes ridehailing's dual role as a complement and substitute to public transit by comparing accessibility levels by both transport modes. Using Uber trip data in six cities in the United States and Europe, they found that Uber is used both in competing and complementary circumstances, even though the impact of ride-hailing on overall service accessibility varies greatly within and between cities. Despite a thorough comparison between accessibility by ride-hailing and transit, Cats et al. (2022) do not analyze how ridehailing could expand transit accessibility when used as a first-mile service for transit. Nonetheless, this extended analysis was done by Abdelwahab et al. (2021), who used Uber trip data in Toronto to examine the extent to which ride-hailing helps advance transportation equity in terms of employment accessibility. The authors have found that, overall, ride-hailing outperforms transit in providing accessibility but integrating ride-hailing with public transit does little to improve access to jobs, particularly in deprived neighborhoods.

There is an important methodological characteristic in common in the works of Cats et al. (2022) and Abdelwahab et al. (2021). Both studies consider only the travel time of the fastest transit route between each origin-destination pair and then calculate the fare for those journeys sequentially, to later estimate accessibility using a generalized travel cost function that combines time and money into a single cost value. This is also how generalized travel costs have been previously used in several studies in the accessibility literature (Bocarejo & Oviedo, 2012; El-Geneidy et al., 2016; Geurs et al., 2010). However, this approach can be problematic for studies that try to account for out-ofpocket costs when analyzing accessibility by transit and ride-hailing for at least two reasons.

First, the generalized travel cost approach does not account for tradeoffs between travel time and monetary costs. Oftentimes, multiple transit paths exist between a given origin-destination pair, and some path alternatives may involve longer travel times at cheaper costs, while others may be faster at a higher cost (for example with premium express services or trips that involve transfers between modes). There is no clear answer as to which path should be considered, given the variety of circumstances under which individuals have different preferences based on their monetary and time constraints. In fact, considering only one of the alternatives poses important implications for transportation equity analyses. On one hand, considering only the fastest paths may result in prohibitive costs for individuals with stricter financial constraints, as these paths may be more expensive due to transfers or faster modes. On the other hand, considering a single cheaper path may bias accessibility estimates by ignoring more efficient travel alternatives. In both cases, accessibility estimates may be biased and over or underestimated for individuals who are more money- or time-constrained in their travel choices, which often includes poorer and wealthier groups, respectively. Therefore, ignoring the trade-offs between travel time and financial costs can be particularly problematic for studies concerned with equity issues

Secondly, calculating accessibility based on a generalized travel cost function poses several challenges. One issue is the conversion of money to time (or vice versa) by assigning a monetary value to time. This process involves several ad-hoc methodological decisions that may not be appropriate for studies that are concerned with transportation equity (Martens & Di Ciommo, 2017). Assigning a value of time (VOT) that varies according to people's income levels, for example, implies that the time of wealthier individuals is worth more than that of poorer citizens, which can perpetuate historical privilege (Börjesson & Eliasson, 2019; Goodwin, 1974). An alternative approach is to use a fixed VOT, which partially addresses this issue. However, it is not clear what value should be used, leading to ad-hoc decisions that compromise the comparability of different studies and may introduce other forms of bias. Additionally, converting an absolute amount of money to time with a fixed VOT is relatively straightforward, but doing so when looking at trip monetary cost as a share of one's total budget is much more complex and reduces the communicability and interpretability of results.

To date, the majority of research on public transit accessibility has primarily focused on travel time due to the limitations of routing engines, such as OpenTripPlanner and ArcGIS Network Analyst, which overlook the cost of transit fares. Only more recently, a new routing algorithm with a multiobjective optimization method has been developed that simultaneously accounts for time and cost constraints in public transit accessibility routing (Conway & Stewart, 2019). This method is utilized in the present study, allowing us to consider a large number of trip alternatives between the same origin-destination pair, ranging from faster though expensive itineraries to slower and cheaper routes. The method is further explained in the methods section.

3. Study area

Rio de Janeiro is the second largest city of Brazil and the fifth largest in Latin America with approximately 6.7 million inhabitants, making it one of the largest markets for ride-hailing in the continent. In Rio, ridehailing services were used by approximately 6 % of the population that used some form of motorized transportation in 2018, twice as much as the national average of 3.1 % in all Brazilian cities (Warwar & Pereira, 2022). While there is no available data with information on the purpose of ride-hailing trips in Rio de Janeiro, the Uber trip data set used in this paper (see Section 4) indicates that approximately 12 % of all Uber trips in Rio are taken during the morning peak period, when most trips are often work-related or linked to other regular activities. Ride-hailing users in the city make on average 8 trips per month, at an average cost of approximately 33 BRL per trip (Warwar & Pereira, 2022). Ride-hailing users in Brazil are largely made up of women (over 60%), whites (approximately 48%), as well as young (46% are between 15 and 35 years old) and wealthier individuals (ibid.). In Rio, in particular, 60% of ride-hailing users are among the 40% wealthiest in the city, while only 23% of users are among the city's poorest 40%.

The spatial distributions of population, income and employment opportunities in Rio are presented in (Fig. 1). Wealthier population groups generally reside in the city's southern and southeastern regions (Fig. 1A), which are among the most densely populated areas in the city (Fig. 1B). High-income neighborhoods also tend to be relatively closer to the CBD, where the majority of jobs are located (Fig. 1C), thus facilitating the access to many urban facilities, which can be reached from such neighborhoods with a short ride. The poorest population, conversely, is mainly located in the northern and western regions of the city, which are less densely populated and very far from the city's main employment hubs, making ride-hailing more expensive and, thus, less affordable.

The public transportation network in Rio de Janeiro also tends to serve the population unequally (Fig. 1D). The subway system, usually regarded as having better quality than the other transit alternatives, serves mainly the wealthiest areas in the southeastern region and the CBD, although it also stretches to some neighborhoods in the northern region of the city. Rail and BRT services, which are generally slower and more prone to service disruptions than the subway, provide transportation over longer distances to the poorest areas located in the western- and northern-most ends of the city. The city also counts on a light rail system that runs mainly through the CBD and its vicinities, a ferry system that connects Rio to adjacent cities and a widespread municipal bus system that shares right-of-way with automobiles, thus leading to slower operational speeds. The buses are highly used by lowincome people because they are less expensive than higher-order transit alternatives in Rio (more details in Section 4). The transit system's spatial configuration coupled with the co-distribution of people and jobs in the city leads to a scenario where those who depend on public transport the most, often face the longest commuting times. As per Rio's latest travel survey, 17 % of the transit trips conducted in the city were longer than 90 min, with an average commuting time of 57 min (Central, 2016).

4. Data

We combine data from many different sources. Socioeconomic and population data come from the 2010 Population Census (IBGE, 2011), the latest census available in Brazil. Data on job locations come from the 2019 RAIS, an administrative records product developed by the Ministry of Economy with information on all formal workers in the country.¹ These datasets are spatially aggregated to a hexagonal grid based on Uber's H3 index (Brodsky, 2018) at resolution 8, in which each cell has a radius of approximately 461 m and area of 0.74 Km². Income data were adjusted based on the inflation from 2010 to 2019 using the official IPCA index (Broad Consumer Price Index, in Portuguese).

The street network and the pedestrian infrastructure were extracted from OpenStreetMap in August 2019. To factor in street slopes when

¹ Although there is no recent data available on the location of informal jobs, these jobs are relatively more accessible with shorter commute times and distances than formal jobs (Motte-Baumvol et al., 2016). Moreover, the 2003 household travel survey of Rio suggests a very similar spatial distribution of both types of jobs in Rio. These data indicates that the number of formal and informal jobs in the same zone is highly correlated (0.78 Pearson correlation statistically significant at 0.001), and that the number of formal jobs in neighboring zones (Global Bivariate Moran's I of 0.30 statistically significant at 0.001).



Fig. 1. Spatial distribution of (A) deciles of income per capita, (B) population density, (C) job density, and (D) main transit routes in Rio de Janeiro. Obs. Transit network from 2019, while population and income data are from 2010 adjusted by inflation between 2010 and 2019.

calculating walk travel times, we used elevation data with a spatial resolution of 30 m, produced by the Shuttle Radar Topography Mission (Farr et al., 2007). Data describing the public transport system in October 2019 was provided by Fetranspor (Rio state's federation of passenger transport companies) and SuperVia (train system operator) in the General Transit Feed Specification (GTFS) format. Intermunicipal buses were removed from the GTFS because we could not find reliable information on the fare costs of these services. Since we only analyze accessibility within the municipality of Rio de Janeiro, and because intermunicipal buses are more expensive than the other available transit modes in the city, removing these services does not significantly affect our results and conclusions.

Information of public transit fares were collected from Rio's transport authority website.² The fare prices in effect in 2019 are summarized in Table 1. Single mode fares are not dramatically different, a subway ticket is only 25 % more expensive than a municipal bus ride, for example. Nonetheless, the lack of fare integration between specific modes significantly affects the affordability of multimodal trips. In particular, the lack of integration between the municipal buses and the subway and the small discount in transfers between the buses and the rail makes faster transit options more expensive for low-income travelers who live far from rapid transit corridors and need to connect to them via buses.

To estimate accessibility by ride-hailing, we used two data sets provided by Uber under a non-disclosure agreement. The first data set is a travel matrix table that covers 152 million Uber trips taken from March

Table 1

Transit fares in effect in Rio de Janeiro in 2019.

Single mode		Trip integration	
Mode	Fare (BRL ^a)	Integration	Fare (BRL)
Light-rail (VLT)	3.80	VLT + VLT	3.80
Municipal bus	4.05	Municipal bus + municipal bus	4.05
BRT	4.05	Municipal bus + VLT	4.05
Rail	4.7	Municipal bus + BRT	4.05
Subway	5.00	BRT + subway	7.10
Ferry	6.30	Rail + subway	8.55
		Bus/BRT + rail (no discounts)	8.75
		Bus + subway (no discounts)	9.05

^a obs. As of October 2019, 1000 BRL (Brazilian reais) was worth approximately 241 USD.

8th to December 20th, 2019. This data set consists of aggregate information on the total number of trips between origin-destination pairs, as well as their average distance, speed and fare. The data is spatially aggregated over a hexagonal grid based on the H3 index using resolution 8. This information is aggregated by day of the week (either weekdays or weekends) and by time blocks (morning peak 6 am - 9 am, day off-peak 9 am - 5 pm, evening peak 5 pm - 8 pm or night off-peak 8 pm - 6 am). In order to calculate employment accessibility considering the conditions most transit users would face when commuting to work, we used Uber trip data collected on weekdays during the morning peak. Due to privacy concerns, the travel matrix data provided by Uber only includes information for origin-destination pairs between which at least 10 trips were taken, after accounting for the aggregations listed above. To fill the data gaps between origins and destinations whose trip count was lower than

² Available at <u>https://www.cartaoriocard.com.br/rcc/institucional/tarifas.</u>

10, we used Streetmap Premium, a commercial data set licensed by ESRI that brings historical traffic speed information based on GPS data collected multiple times of the day between the first quarters of 2018 and 2020.

The second data set provided by Uber presents information on the spatial distribution of pickup wait times (average, 25th, 50th and 75th percentiles) and number of pickups in Rio de Janeiro. The data set is spatially aggregated into hexagonal cells of different sizes (H3 index resolutions 8 and 9) to guarantee a minimum of 10 pickups per cell, for data fidelity and privacy concerns. This data set is also aggregated by day of the week and time blocks.

5. Methods

To investigate how ride-hailing services impact employment accessibility both as a standalone transport mode and when used in conjunction with public transport, we followed a method composed of four steps. The first involved calculating travel time and monetary cost matrices by public transit. The second step was to calculate a similar matrix, but for ride-hailing services. Next, we combined the matrices from the previous steps to generate a third travel matrix, in which ride-hailing is used as a first-mile feeder to rapid transit stations and the rest of the trip is completed by transit services. Finally, in the fourth step we calculated accessibility levels using each one of these matrices and compared how the accessibility conditions of each scenario are distributed in space and across different socioeconomic groups. Each methodological step is detailed below. In all of these steps, the spatial units of analysis are hexagonal cells from the H3 spatial grid (Brodsky, 2018) at resolution 8.

Data processing and visualization were conducted in R. The code used in this paper is publicly available in a GitHub repository,³ and can be used as a reference when replicating the method of this paper in other contexts.

5.1. Travel time and cost matrices by public transit

Calculating the travel matrix between origins and destinations is a fundamental step to estimate accessibility levels in a given area. However, simultaneously accounting for both travel time and monetary cost in multimodal transport networks raises two challenges: the first is that the journey between a given origin-destination pair may be completed by multiple trip alternatives that may not be dominated by a supposedly "optimal" alternative. For example, the journey between one point to another might be completed with a fast but expensive subway trip (e.g. a 15-minute trip that costs 10 BRL) OR with a longer though cheaper bus trip (e.g. a 40-minute trip that costs 5 BRL). The first challenge is which path alternative should be considered when calculating accessibility estimates. Passengers looking to save time may prefer the first alternative, while those looking to save money might opt for the second. If we had omitted the first trip alternative from the travel matrix, we would have mistakenly concluded that the destination was inaccessible for trips that cost less than 10 BRL when, in fact, this is not the case. Similarly, if the second alternative had been omitted, we would have wrongly assumed that the destination was inaccessible for journeys of less than 40 min, which is also untrue. To overcome this issue, one needs to consider the full set of trip alternatives when calculating accessibility estimates, requiring the routing engine to output these alternatives in the first place - which takes us to the second challenge.

The second challenge is that performing multimodal transit routing with multiobjective optimization is a complex and computing-intensive task (Delling et al., 2015), specially because transit fares are often pathdependent; i.e. the fare cost of a ride can change depending on transfers that may have happened earlier in the journey, based on the fare system rules (Conway & Stewart, 2019; Lo et al., 2003). In a recent work, Conway and Stewart (2019) introduced a novel multiobjective optimization routing method to account for cost constraints in public transit accessibility metrics. Rather than finding only the fastest trip alternative regardless of monetary cost, the method is able to determine for a given origin-destination pair the multiple journey alternatives that are optimal in terms of both travel time and cost. As such, the method returns the set of alternative paths that form a frontier beyond which no journey is both faster and cheaper, ignoring trips that are simultaneously more expensive and slower than any existing alternatives. In other words, the model finds the fastest trip for each possible combination of fares in the transit network. The routing model proposed by the authors was implemented in the R5 routing engine developed by Conveyal.

In this paper, we used Conway & Stewart's (2019) routing model, available in R through the r5r package (Pereira et al., 2021), to generate the transit Pareto frontier for each OD pair. These frontiers include for each origin-destination pair both walk-only trips and trips in which transit is the main mode and the access to and egress from stations is done by foot. We have considered a walk speed of 3.6 km/h, a maximum walk time of 30 min to access/egress from transit stops and that a trip would consist of at most 4 transit legs. We have also considered trips departing every minute from 7 am to 8 am, and used the median travel time within this time window to calculate the accessibility levels. This strategy helps us mitigate the impact of statistical noises related to the variation of transit services availability within the morning peak (Conway et al., 2018). The travel time of a trip consists of the duration of a door-to-door journey, including the time it takes to walk from the origin to the departure stop, any wait and transfer time that occurs at stops, invehicle time and the time it takes to walk from the arrival stop to the destination. The monetary cost of a trip is the sum of the fares paid in each of its legs, including any fare integration discounts available when using the Riocard Mais smartcard as presented in Table 1.

5.2. Travel time and cost matrices by ride-hailing

A ride-hailing travel matrix covering all possible origin-destination pairs is essential to calculate the accessibility conditions that result from using ride-hailing services, both as a standalone mode and as a first-mile feeder to transit. However, the aggregate travel matrix data provided by Uber does not cover all possible combinations of origindestination pairs. To fill in the data gaps, we used a multi-stage process.

First, we calculated travel times and distances by automobile during the morning peak for all possible origin-destination pairs using the Streetmap Premium data and the Network Analyst plugin in ArcGIS Pro. Using the origin-destination pairs for which we had Uber trip data, we used a linear regression model to predict Uber travel times and distances based on automobile travel times and distances calculated with Network Analyst (R^2 fit of 0.82 and 0.95, respectively). We used the coefficients derived from this model to estimate Uber's trips distances and travel times for pairs not covered in Uber's original data set. In a second stage, we used a similar regression to predict the cost of Uber trips based on travel time and distance (R^2 fit of 0.96), and used these coefficients to estimate Uber's trip costs for pairs not covered in Uber's data set. We manually imputed the value of 5 BRL for the cases where the predicted trip cost was below 5 BRL, which was the minimum Uber fare in Rio de Janeiro in 2019.

The travel times listed in Uber's data set and estimated through the linear regression include only in-vehicle time. To properly calculate the duration of a door-to-door journey we have to include the waiting time at the beginning of the trip (from the trip request to the vehicle arrival at the origin) to the total travel time. To do this, we summed the average waiting time at each origin, listed in the pickups data set provided by Uber, to the in-vehicle times between each origin-destination pair. This is a crucial step when estimating accessibility by ride-hailing, because areas under-supplied by drivers will have, on average, higher waiting times - thus potentially lower accessibility levels. For the sake of

³ Available at <<<u>https://github.com/ipeaGIT/access_uber>></u>.

simplicity, the calculation of total travel time assumes there is no walking before boarding and after arrival. For hexagons with missing data on waiting time due to privacy reasons, we estimated their waiting time as the average waiting time of their neighbors.

Finally, we have also included potential walk-only trips to the ridehailing travel matrix, which is important to make sure we calculate accessibility levels correctly. For example, if we estimate accessibility with a monetary threshold that prevents any ride-hailing trips to be taken, but with a temporal threshold that allows some walking trips to be completed, we still have to consider the opportunities that can be accessed by foot from the origins, like one does when calculating transit accessibility. To do this, we have calculated a walk travel matrix with r5r and created a larger matrix between all combinations of origins and destinations which include both walking-only trips and Uber trips. The result is a Pareto frontier that considers walk-only and Uber-only trips.

5.3. Travel time and cost matrices considering first-mile by ride-hailing

The next step in our method was to calculate travel matrices considering a first-mile leg by ride-hailing and the remainder of the trip done by transit. To calculate the travel matrix by ride-hailing combined with transit we followed a four-step method. First, we calculated the time and cost by Uber from each origin to all hexagons containing rapid transit stations. We then merged this data set to the Pareto frontier composed of trips departing from each rapid transit station to all destinations in the city. Here, we had to consider multiple departure times by transit to accommodate the transfer between ride-hailing and transit. For example, if the ride-hailing trip arrived at the subway station at 7:15 am, we had to merge it with transit trips that departed from the same subway station after 7:15 am.

Next, we calculated the total travel time and monetary cost of each trip by adding the travel times and monetary costs of the first-mile leg by ride-hailing and the transit leg. In cases where multiple trip alternatives existed between an origin-destination pair (e.g. one could go from point A to B either via transit stations X or Y), we kept only the alternatives that were simultaneously more efficient both in terms of travel time and out-of-pocket cost than the other options. In other words, we calculated a Pareto frontier composed of trips whose first-mile was traveled by ride-hailing and the rest by rapid transit.

Finally, we merged both the Pareto frontier by ride-hailing combined with mass transit and the Pareto frontier by transit into a single larger frontier. As a result, the final Pareto frontier by ride-hailing combined with transit also includes transit trips whose access to stations was completed by foot, as long as these alternatives are not simultaneously slower and more expensive than the trips that include ride-hailing as a first-mile mode.

5.4. Accessibility estimates

To calculate accessibility levels, we used cumulative opportunity measures simultaneously considering travel time and monetary costs thresholds. Accessibility levels are then calculated in two distinct ways. First, we considered absolute monetary costs, as described in Eqs. (1), (1.1) and (1.2).

$$A_{i} = \sum_{j=1}^{n} O_{j} \times max_{k \in K} \left(f(t_{ijk}) \times g(c_{ijk}) \right)$$

$$\tag{1}$$

$$f(t_{ijk}) = 1, iff(t_{ijk}) \le T; 0, iff(t_{ijk}) > T$$

$$(1.1)$$

$$g(c_{ijk}) = 1, ifg(c_{ijk}) \le C; 0, ifg(c_{ijk}) > C$$

$$(1.2)$$

where A_i is the accessibility to jobs at origin *i*, O_j is the number of jobs at destination *j*, *K* is the total set of paths between origin *i* and destination *j*, t_{ijk} is the travel time of path *k* between origin *i* and destination *j*, *T* is the travel time threshold, $f(t_{ijk})$ is a binary function which returns values 0 or 1 based on the travel time, c_{ijk} is the absolute monetary cost of path *k*

between origin *i* and destination *j*, *C* is the absolute monetary cost threshold, and $g(c_{ijk})$ is a binary function which returns values 0 or 1 based on the absolute monetary cost between the origin *i* and destination *j*.

Because the burden of transportation costs can weigh differently for low- and high-income individuals, we also calculated accessibility considering monetary costs relative to the income per capita at the trip origin, as described in Eqs. (2), (2.1) and (2.2). This was done considering what would be the total monthly commute cost of an individual who uses the same path to commute to and from work every business day as a share of her/his monthly income, calculated as the income per capita before taxes of the trip origin hexagonal cell.⁴ Assuming 22 business days in a month, the relative monthly cost of a trip is given by Eq. (2.2).

$$A_{i} = \sum_{j=1}^{n} O_{j} \times max_{k \in K} \left(f(t_{ijk}) \times h(b_{ijk}) \right)$$
(2)

$$h(b_{ijk}) = 1, ifh(b_{ijk}) \le B; 0, ifh(b_{ijk}) > B$$

$$(2.1)$$

$$b_{ijk} = \frac{c_{ijk} \times 2 \times 22}{I_i} \tag{2.2}$$

where $h(b_{ijk})$ is a binary function which returns values 0 or 1 based on the relative monthly cost, b_{ijk} is the relative monthly cost of a trip with path k between origin i and destination j in terms of one's monthly budget, B is the relative monthly cost threshold, and I_i is the monthly income per capita at origin i.

The main advantages of cumulative opportunities measures are that they are easy to communicate, operationalize and interpret (Geurs & van Wee, 2004). However, they are often criticized for the need to set arbitrary cost thresholds, for ignoring any activities outside these limits and for the fact that all activities within the thresholds are considered equally reachable (Geurs & van Wee, 2004; Pereira, 2019). In order to mitigate this issue, we have used several combinations of thresholds selected from three different distributions: travel times cutoffs range from 1 to 90 min, every 1 min (1, 2, 3, ..., 88, 89, 90); absolute monetary costs cutoffs range from 0 BRL to 24 BRL, every 0.05 BRL (BRL 0, 0.05, 0.10, ..., 23.90, 23.95, 24); relative monetary costs cutoffs range from 0 % to 40 %, every 1 % (0 %, 1 %, 2 %, ..., 38 %, 39 %, 40 %).

For the sake of brevity, some of the results presented in the paper are based on selected thresholds. When focusing on a single time threshold, we have chosen to use a travel time of 60 min, which captures the results for moderately long trips and which is close to the average transit commute time in Rio (57 min) as per the last travel survey (Central, 2016). The chosen absolute monetary costs threshold were 6 BRL, 12 BRL, 18 BRL and 24 BRL. These cutoffs cover potential transit and ridehailing trips with distinct lengths and costs, ranging from just above the minimum Uber fare in Rio (5 BRL) to moderately expensive trips (24 BRL). Finally, the chosen relative monetary costs thresholds were 10 %, 20 %, 30 % and 40 %, highlighting the results when considering different affordability thresholds.

6. Results

Considering the Pareto frontier of travel time and monetary cost for all origin-destination pairs in Rio allows us to calculate the average

⁴ Calculating relative monthly costs as a percentage of the income per capita at a given region means that we have assumed that the income is evenly shared by every member of a household. In practice, however, the total income of a household is not evenly shared by its individuals, with some household members requiring larger budgets to perform their activities, such as commuting. Data availability prevents us from calculating relative monthly costs considering household sizes and the number of economically active household members, which is why we opted to use the first method.

number of accessible jobs in the city given every possible combination of time and money thresholds. The result of this operation is a Pareto surface that represents the Pareto-optimal accessibility levels in the city i.e. the number of accessible jobs calculated from the full set of trips that optimize for time and money combinations between each origin-destination pair (Fig. 2).

The transit-only panel shows that average accessibility levels by transit are largely influenced by travel time once the passenger is able to spend enough to afford a bus ticket, around 4 BRL. Above this value, though, average accessibility levels are almost unaffected by more expensive transit tickets, even though they would allow one to ride faster modes, such as the rail and the subway. The exception to this can be seen around the 8 and 9 BRL marks, values with which the passengers can make transfers between transport modes, such as the municipal buses, the subway and the rail. The dominance of travel time cutoffs over monetary cost cutoffs on shaping the accessibility levels happens largely because Rio uses a flat fare system, in which prices remain constant regardless of distance/time traveled. The ride-hailing-only panel, on the other hand, shows that money exerts a stronger influence than time on the accessibility levels by this transport mode: the more expensive the ride-hailing trip, the farther one can go with it. It also shows that the accessibility benefits from ride-hailing-only trips become more salient at higher costs, above 15 BRL, whereas transit-only accessibility benefits are significant even when considering cheap 5 BRL trips. Finally, the figure also shows how using ride-hailing as a first mile connection to transit can significantly expand employment accessibility for transit users. The marginal accessibility gains for every additional BRL becomes much more pronounced than on the transit-only scenario, while accessibility levels at low monetary thresholds remain much higher than when considering ride-hailing-only trips.

Although the Pareto surface can help us visualize the marginal impacts of time and money on average accessibility levels, it makes it hard to analyze and compare accessibility conditions at particular monetary and temporal thresholds. Thus, to facilitate the interpretation and comparison of results, in the rest of the paper we report our findings using selected combinations of travel time and monetary thresholds.

6.1. Spatial distribution of accessibility

Fig. 3 shows the spatial distribution of employment accessibility by transit-only, ride-hailing-only, and ride-hailing combined with transit in up to 60 min of travel and multiple absolute monetary cost thresholds. This figure presents the spatial context for the analyses in this paper, highlighting how accessibility conditions are distributed across the city in each transport mode scenario. Markedly, the maps show a sharp East-West divide in all three scenarios, in which residents of the east side of the city have higher accessibility levels. This pattern can be explained by the large concentration of jobs and public transit services near the city center, located in the East side of Rio (see Fig. 1).

The maps in Fig. 3 also show significant differences in the accessibility levels provided by the different transport mode alternatives. The transit system provides relatively high accessibility levels along mass transit corridors that run from the CBD towards the west, even at low monetary costs when compared to ride-hailing. A 6 BRL budget is enough to pay for a transit trip by bus (with transfers allowed between buses and BRTs), by train or by subway, without any travel time limit. Compared to the ride-hailing as a standalone mode, the transit system configuration results in higher accessibility in areas farther from the CBD, showing that, in Rio, mass transit can serve population groups that live in the outskirts of the city at relatively low costs. Nonetheless, very little additional accessibility is gained by spending more money on transit alone when compared to the other alternatives.

For ride-hailing services, on the other hand, accessibility is severely impacted by monetary costs restrictions. A 6 BRL budget only allows one to access the jobs in the immediate vicinity of the trip origin, resulting in very low accessibility levels on average. The number of accessible jobs consistently increases when taking more expensive trips into account. Nonetheless, the fact that accessibility is still largely concentrated near the city center, even at higher cost thresholds, indicates that ride-hailing as a standalone mode is very limited in providing access when costs are capped to 24 BRL.

The Fig. 3 also shows how the spatial distribution of accessibility levels are affected when allowing passengers to combine ride-hailing and public transit. Ride-hailing can extend the reach of the transit system far beyond the immediate vicinity of transit stations when used as an on-demand feeder service to the main transportation system. However, the accessibility gains that result from this combination are conditioned to higher monetary spending. Accessibility levels remain unchanged when allowing ride-hailing connections to public transport with a 6 BRL budget, but higher budgets result in accessibility gains in areas around the main transit corridors, from where ride-hailing services potentially substitute slow feeder transit trips to these corridors and long walks to the stations.

6.2. Average accessibility levels by transport mode and income group

The maps presented in Fig. 3 present a snapshot of the accessibility distribution when limiting trips to a given travel time threshold (60 min). The arbitrary choice of thresholds, however, may severely affect the results and conclusions of accessibility analyses (Pereira, 2019), an issue that we try to mitigate by considering several different travel time cut-offs in our accessibility analyses. Fig. 4 shows how average accessibility levels vary by transport mode with different combinations of absolute monetary cost and travel time thresholds.

The figure reasserts that a 6 BRL budget is not enough to generate accessibility gains from ride-hailing, neither as a standalone mode nor as a first-mile mode to transit. When using it as a standalone transport mode, only very short distances can be traveled when spending that amount of money. At the same time, it is virtually impossible to use ride-hailing as a first-mile mode to transit spending only 6 BRL. This is because the minimum Uber fare in Rio (5 BRL) plus the cheapest rapid transit fare (4.05 BRL) entails trips that are more expensive than the selected threshold. As a result, the curve for ride-hailing only barely rises above 0 % average accessibility, while the transit-only and the ride-hailing combined with transit curves overlap each other (only the latter is shown in the figure).

The advantage of ride-hailing becomes clearer with higher budgets (12 BRL onwards). In such cases, the ride-hailing as a standalone mode is able to provide much higher accessibility levels than the public transport system when considering relatively short trips, as a result of its higher speeds and lower access times. For example, considering a travel time threshold of 30 min and monetary cost thresholds of 18 and 24 BRL, the average accessibility by ride-hailing was 5 and 7 times higher than by transit, respectively. It is in these relatively short trips where ridehailing has higher competitive advantage over transit. This advantage is limited to shorter trips due to cost constraints: the cost of ride-hailing trips, as opposed to the cost of transit trips in Rio, increases with its distance, making long trips more expensive than the monetary thresholds considered in the analysis. Consequently, the accessibility that results from using ride-hailing as a standalone mode reaches a ceiling at relatively low temporal thresholds, whereas the transit accessibility keeps growing for long travel times.

The advantages of combining ride-hailing with transit, on the other hand, become more evident as the time threshold increases. From the 30 min mark onward, approximately, the accessibility curve of ride-hailing combined with transit curve (Fig. 4) detaches from, and remains consistently higher than, the transit-only curve. The accessibility benefits of using ride-hailing as a first-mile feeder to transit also become more prominent when taking into account more expensive trips, as shown by the wider gap between the ride-hailing combined with transit and the transit-only curves. Considering a temporal threshold of 60 min, for example, the use of ride-hailing as a first-mile feeder to transit expands



Fig. 2. Pareto-optimal accessibility with the average number of jobs accessible for each combination of travel time and monetary costs.



0% 16% 32% 48%

Fig. 3. Spatial distribution of employment accessibility by different transport mode alternatives in under 60 min of total travel time and multiple monetary cost thresholds. Rio de Janeiro, 2019.

the average employment accessibility of transit riders by approximately 61~% and 75~% when taking into account trips of up to 18~BRL and 24~BRL, respectively.

The previous figures show how absolute monetary costs impact

potential employment accessibility levels, but they do not account for the affordability of these costs according to travelers' incomes. The costs of daily trips, however, may impose significant financial burdens on commuters' budget, especially in a city like Rio, with high poverty levels:



Fig. 4. Average employment accessibility for different combinations of travel time and absolute monetary cost thresholds by transport mode. Rio de Janeiro, 2019.

wealthy and poor passengers may pay the same to travel from one place to another, but the former spends a much smaller share of their budget than the latter. Therefore, using ride-hailing to commute to and from work, be it as a standalone mode or as a first-mile feeder to public transport, may not be viable for a large part of the population due to financial constraints. This makes it particularly important to assess the accessibility conditions considering monetary restrictions not only in absolute terms, but also as a share of one's income.

Fig. 5 shows how the average accessibility levels in Rio de Janeiro vary by transport mode with different combinations of relative monthly cost and travel time thresholds. Because aggregate analyses may mask significant differences among population groups, the figure exhibits the



Fig. 5. Average employment accessibility for high- and low-income groups for different combinations of travel time and monetary costs relative to income by transport mode. Rio de Janeiro, 2019.

average accessibility levels of the wealthiest 10 % and the poorest 40 % of the population separately.⁵ The different scenarios of transport mode are indicated by the color of each curve, while the line type (whether solid or dashed) indicates the population group that each curve refers to.

The accessibility levels of the wealthiest, shown in solid lines, follow a similar pattern to the average accessibility levels of the entire population, but at much higher levels: with public transport trips of up to 90 min and that may compromise up to 40 % of their income, they can access approximately 76 % of all jobs in the city. As hypothesized before, using ride-hailing services, both as a standalone and as a first-mile mode, yields much larger accessibility benefits to the wealthiest citizens than to the population as a whole. The wealthiest residents can ride more expensive and longer ride-hailing trips than the rest of the population when compromising the same share of their income on transport, thus significantly increasing their capability to access jobs within a given time threshold.

The accessibility conditions of the poorest, shown in dashed lines, on the other hand, are remarkably different. Allowing transport expenditures of up to 20 % of their monthly income barely results in any accessibility benefits, which only become more noticeable when considering trips that compromise 30 % of their income on commuting costs. Another important difference is that the ride-hailing combined with transit and the transit-only curves never separate from each other, resulting in only the latter being visible in the chart. Thus, using a ridehailing service as a first-mile mode to transit does not yield any employment accessibility gains to low-income people. In fact, Rio's poorest residents can barely afford public transport itself, making a ridehailing leg to reach transit prohibitively expensive. Using ride-hailing as a standalone mode is also impractical since Uber's base fare is more expensive than most transit fares and trip costs increase too quickly with trips distances.

6.3. Accessibility gains from integrating ride-hailing and transit

Now we turn to the potential accessibility gains from using ridehailing as a first-mile leg to mass transit and how such benefits are distributed both in space and across the population. Fig. 6 shows how many more jobs one could reach in 60 min from different areas of Rio by combining ride-hailing and transit compared to riding transit-only trips.

Considering absolute monetary costs (Fig. 6A), significant accessibility gains can be obtained throughout the city when considering trips that cost up to 12 BRL. These gains are more pronounced along train and BRT corridors, ranging from the north of the city towards the west, and very limited at the southern and southeastern regions of the city, where some of the wealthier neighborhoods are located. In these wealthier neighborhoods, the accessibility benefits that result from combining ride-hailing and transit only become more prominent when considering more expensive trips, of up to 18 and 24 BRL. This is explained by the fact that this area has relatively high accessibility levels even without considering ride-hailing services, as it is well served by public transport, especially the subway. The accessibility gains that result from using ridehailing as a first-mile mode to rapid transit are, therefore, relatively well distributed across the city when calculated using absolute monetary cost thresholds. The locations that benefit the most are concentrated along rapid transit corridors, but significant gains can be seen both on poorer neighborhoods located towards the western and northern regions of the city and on wealthy areas located at the south and southeastern regions.

When transport affordability is considered (Fig. 6B), however, the spatial distribution of accessibility gains changes dramatically. Accessibility gains become largely concentrated in high-income

neighborhoods in the south region and near the CBD, whereas lowincome neighborhoods that could potentially gain large accessibility sums when calculating it with absolute cost thresholds show barely any gains when calculating it with relative thresholds. This discrepancy is a result of the low purchasing power of the residents of such regions, who cannot afford the costs of first-mile ride-hailing trips.

To demonstrate how accessibility differences across income groups are affected by these spatial distributions, Fig. 7 presents the distribution of accessibility gains among high- and low-income individuals (the wealthiest 10 % and the poorest 40 %, respectively). Fig. 7A shows that low-income communities could potentially get large accessibility benefits from ride-hailing services, even when compared to those accrued by high-income groups: on average, accessibility gains are higher for the poorest population than for the richest when limiting monetary costs to 12 BRL (with a median gain of 2 % versus 0 %), and very similar when limiting costs to 18 and 24 BRL (with median gains of 6 % and 8 %, respectively). However, because of affordability constraints, the accessibility gains of low-income individuals are nearly eliminated (Fig. 7B), as opposed to those of high-income groups, suggesting that on-demand rideshare or microtransit services with dynamic routing could largely benefit low-income neighborhoods if financial barriers were alleviated.

In summary, our findings illustrate how differences in accessibility between population groups in Rio de Janeiro are largely shaped by the high socioeconomic and spatial inequalities observed in the city. Even though there are significant accessibility gains to be obtained by integrating ride-hailing services to the transit system in Rio, there are important financial barriers that limit who can benefit from these services on a daily-basis. As a result, ride-hailing becomes a limited mobility solution to improve mass transit and reduce social exclusion particularly in the Global South context where many cities face high poverty rates.

7. Conclusion

This study examined how ride-hailing shapes spatial and socioeconomic differences in employment accessibility in Rio de Janeiro while accounting for the trade-off between travel time and monetary costs, considering ride-hailing both as a standalone transportation mode and when combined with transit as a first-mile feeder. Our results indicate that ride-hailing provides higher employment accessibility than public transit when considering trips of up to 40 min, and that the use of ridehailing as a first-mile feeder to transit can significantly expand employment accessibility when considering trips that take at least 30 min. In both cases, though, the accessibility gains from ride-hailing come at relatively high out-of-pocket costs. These gains become more pronounced when considering trip costs of 12 BRL or more, which are expensive when compared to transit fares. As a result, when we account for different affordability thresholds where a person could only allocate between 10 % and 40 % of their monthly income on commuting costs, the accessibility benefits of ride-hailing services accrue mostly to highincome individuals. Because transit fare prices in Rio de Janeiro are fixed regardless of traveled distances, transit becomes substantially cheaper than ride-hailing and thus better able to cater to the needs of low-income populations, most of whom live in urban peripheries and engage in long commutes.

These results show that ride-hailing services can importantly shape access to opportunities in cities. However, our findings suggest that these services do not provide an equitable alternative to public transit systems - neither as a standalone mode nor when combined with transit. While Transport Network Companies (TNCs) can potentially overcome existing transit network gaps, the potential accessibility benefits from TNCs to transit-dependent populations are rather limited, since these are mostly composed of low-income individuals for whom monetary costs are a critical barrier. This suggests that the growth of market-priced TNCs is likely to increase levels of transport inequalities within Rio.

This paper also shows how accessibility estimates can vary greatly

⁵ The cut-offs of the wealthiest 10 % and poorest 40 % is inspired by the Palma ratio, an inequality measure commonly used in transport studies to compare average accessibility conditions of the most well-off and disadvantaged population groups (Guzman & Oviedo, 2018; Herszenhut et al., 2022).



Fig. 6. Employment accessibility gains by ride-hailing combined with transit compared to transit-only accessibility considering a 60-minute travel time threshold and multiple monetary cutoff values in terms of absolute costs (A) and cost relative to income (B). Rio de Janeiro, 2019.

given different combinations of travel time and monetary cost thresholds. This result draws attention to a broader methodological question about the extent to which accessibility estimates and equity analyses might be sensitive to whether and how researchers incorporate both of these costs in accessibility calculations. Previous research has shown that by ignoring monetary costs researchers tend to overestimate the accessibility of low-income groups, which may lead to underestimated levels of accessibility poverty and inequality (Herszenhut et al., 2022; Liu & Kwan, 2020). However, even the use of travel cost functions that take monetary costs into account can generate biased results if one follows the commonly adopted practice of considering the fastest journeys and then calculating monetary costs sequentially (Conway & Stewart, 2019). This study illustrates how using multiobjectiveoptimization routing to calculate Pareto frontiers could be a promising way to account for trade-offs between time and money in accessibility research. From an equity perspective, taking this trade-off into account is particularly important, especially in contexts with great social inequalities. However, the use of Pareto frontiers to estimate accessibility is still very new, and more research is needed to investigate how tradeoffs between time and monetary cost can be incorporated in other types of accessibility metrics such as gravity-based, utility and person-based measures. We hope the Pareto frontier function implemented into the R package r5r can unlock some of these research possibilities.

From a policy perspective, the findings of this study indicate that major accessibility gains could be achieved by policies that promote an integration between mass transit and some form of on-demand rideshare. Different transport agencies have been exploring partnerships with TNCs to understand the extent to which ridesharing and



Fig. 7. Distribution of employment accessibility gains by ride-hailing combined with transit compared to transit-only accessibility for different income groups, considering a 60-minute travel time threshold and multiple monetary cutoff values in terms of absolute costs (A) and cost relative to income (B). Rio de Janeiro, 2019.

Obs. Wealthiest population was defined as those in the 10 % with the highest income, while the poorest were defined as those among the 40 % with the lowest income.

microtransit with dynamic routing could provide an efficient means to serve neighborhoods with lower population density as first- and lastmile connection and increase accessibility and ridership (Curtis et al., 2019; Schwieterman & Livingston, 2019). However, the findings are also aligned with the result of previous studies (Brown et al., 2021; Palm et al., 2021) that show that policy efforts on integrating rideshare with transit are likely not going to benefit low-income communities without some form of subsidy. This could be done, for example, by using an integrated payment system with fare discounts for transfers between a rideshare and transit trips at particular transit stations.

One limitation of this study is that it uses the latest census data available, which was collected over 10 years ago and which could generate some bias in the results. Nonetheless, we believe this bias should not be strong enough so that it would invalidate the study given that we have corrected income data for inflation in the period, and given that the spatial distribution of socioeconomic classes in Rio have remained fairly stable over the past few decades. This paper also considers only formal jobs. The lack of information on informal jobs is a common issue in low- and middle-income countries. Different data sources, such as point of interests from OpenStreetMap and social media or satellite imagery data could be explored in future research as a proxy for the spatial distribution of informal jobs. Another limitation is that we consider potential first-mile trips of any length. Even though our results did not significantly change when we capped these trips up to 30 min, future studies could arrive at more robust results if they could use travel behavior data to inform how far people are willing to travel the first-mile trip by ride-hailing. Additionally, it would be interesting to compare accessibility levels by ride-hailing and traditional taxis. This analysis was not possible due to the lack of data, but future studies should consider including a scenario of "transit & taxi" to be compared with the scenario of "transit & ride-hailing".

Moreover, this study has only analyzed employment accessibility considering daily commuting costs for single-passenger trips. Given that the costs of ride-hailing could be shared among multiple passengers, analyzing scenarios in which two or more people ride the same trip together would likely yield more favorable results to TNCs. Future studies that focus on access to less frequent activities (such as healthcare or leisure) could also arrive at different results than the ones found in this paper. Further research should also investigate how accessibility by ride-hailing and transit compare in cities where the costs of transit fares depend on zone- or distance-based rules. Finally, future studies could use Pareto frontiers to investigate how travel mode choices are affected by accessibility levels when accounting for travel time and monetary cost trade-offs.

CRediT authorship contribution statement

Rafael H.M. Pereira: Conceptualization, Methodology, Software, Validation, Resources, Data curation, Writing – original draft, Writing – review & editing, Supervision, Project administration, Visualization. **Daniel Herszenhut:** Conceptualization, Methodology, Software, Formal analysis, Writing – original draft, Writing – review & editing, Visualization. **Marcus Saraiva:** Conceptualization, Methodology, Software, Data curation, Writing – original draft. **Steve Farber:** Writing – review & editing.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

The authors do not have permission to share data.

Acknowledgements

This study was conducted under the Technical Cooperation Agreement 03/2021 (SEI process number 03001.000266/2021-73) signed between the Institute for Applied Economic Research (Ipea) and Uber Brazil to promote research collaborations on urban mobility. The agreement does not involve any funding transfer between the organizations. The authors had access to Uber data through a non-disclosure agreement that in no way limited their autonomy over the content of this work.

References

- Abdelwahab, B., Palm, M., Shalaby, A., & Farber, S. (2021). Evaluating the equity implications of ridehailing through a multi-modal accessibility framework. *Journal of Transport Geography*, 95, Article 103147. https://doi.org/10.1016/j. itrangeo.2021.103147
- Alemi, F., Circella, G., Handy, S., & Mokhtarian, P. (2018). What influences travelers to use Uber? Exploring the factors affecting the adoption of on-demand ride services in California. *Travel Behaviour and Society*, 13, 88–104. https://doi.org/10.1016/j. tbs.2018.06.002
- Anderson, M. L., & Davis, L. W. (2021). Uber and alcohol-related traffic fatalities (Vol. No. w29071). National Bureau of Economic Research.
- Barajas, J. M., & Brown, A. (2021). Not minding the gap: Does ride-hailing serve transit deserts? Journal of Transport Geography, 90, Article 102918. https://doi.org/ 10.1016/j.jtrangeo.2020.102918
- Barkley, B., Garr Pacetti, E., & Bailey, L. (2018). A long ride to work: Job access and the potential impact of ride-hailing in the Pittsburgh area. *Co0hmmunity Development Reports*, 20180905. https://www.clevelandfed.org/publications/cd-reports/2018/a lbtn-20180905-a-long-ride-to-work-job-access-and-the-potential-impact.
- Barnes, S. J., Guo, Y., & Borgo, R. (2020). Sharing the air: Transient impacts of ridehailing introduction on pollution in China. *Transportation Research Part D: Transport* and Environment, 86, Article 102434. https://doi.org/10.1016/j.trd.2020.102434
- Barreto, Y., Silveira Neto, R.d. M., & Carazza, L. (2021). Uber and traffic safety: Evidence from Brazilian cities. *Journal of Urban Economics*, 123, Article 103347. https://doi. org/10.1016/j.jue.2021.103347
- Barrios, J. M., Hochberg, Y. V., & Yi, H. (2019). The cost of convenience: Ridehailing and traffic fatalities (SSRN scholarly paper ID 3361227). Social Science Research Network.. https://doi.org/10.2139/ssrn.3361227
- Bedoya-Maya, F., Scholl, L., Sabogal-Cardona, O., & Oviedo, D. (2022). Interactions of transport network companies (TNCs) and public transit in Medellín. Case Studies on Transport Policy, 10(4), 1965–1979. https://doi.org/10.1016/j.cstp.2022.08.011
- Bocarejo, S. J. P., & Oviedo, H. D. R. (2012). Transport accessibility and social inequities: A tool for identification of mobility needs and evaluation of transport investments. *Journal of Transport Geography*, 24, 142–154. https://doi.org/10.1016/j. jtrangeo.2011.12.004
- Börjesson, M., & Eliasson, J. (2019). Should values of time be differentiated? Transport Reviews, 39(3), 357–375. https://doi.org/10.1080/01441647.2018.1480543
- Brodsky, I. (2018, June 27). H3: Uber's hexagonal hierarchical spatial index. Uber Engineering Blog https://eng.uber.com/h3/.
- Brown, A., Manville, M., & Weber, A. (2021). Can mobility on demand bridge the firstlast mile transit gap? Equity implications of Los Angeles' pilot program. *Transportation Research Interdisciplinary Perspectives*, 10, Article 100396. https://doi. org/10.1016/j.trip.2021.100396
- Brown, A., Klein, N. J., Smart, M. J., & Howell, A. (2022). Buying access one trip at a time. Journal of the American Planning Association, 0(0), 1–13. https://doi.org/ 10.1080/01944363.2022.2027262
- Cats, O., Kucharski, R., Danda, S. R., & Yap, M. (2022). Beyond the dichotomy: How ridehailing competes with and complements public transport. *PLoS One*, 17(1), Article e0262496. https://doi.org/10.1371/journal.pone.0262496
- Ceccato, R., & Diana, M. (2021). Substitution and complementarity patterns between traditional transport means and car sharing: A person and trip level analysis. *Transportation*, 48(4), 1523–1540. https://doi.org/10.1007/s11116-018-9901-8
- Central. (2016). Relatório 4 Planejamento e Execução das Pesquisas: Parte 3: Diagnóstico da Situação Atual. In *Companhia estadual de transportes e logística (Central)*. Rio de Janeiro: Governo do Estado do.

A. (2012). Ridesharing in North America: Past, present, and

Cities 144 (2024) 104663

- Chan, N. D., & Shaheen, S. A. (2012). Ridesharing in North America: Past, present, and future. *Transport Reviews*, 32(1), 93–112. https://doi.org/10.1080/ 01441647.2011.621557
- Conner, C. R., Ray, H. M., McCormack, R. M., Dickey, J. S., Parker, S. L., Zhang, X., ... Kitagawa, R. S. (2021). Association of rideshare use with alcohol-associated motor vehicle crash trauma. *JAMA Surgery*, 156(8), 731–738.
- Conway, M. W., & Stewart, A. F. (2019). Getting Charlie off the MTA: A multiobjective optimization method to account for cost constraints in public transit accessibility metrics. *International Journal of Geographical Information Science*, 33(9), 1759–1787. https://doi.org/10.1080/13658816.2019.1605075
- Conway, M. W., Byrd, A., & Van Eggermond, M. (2018). Accounting for uncertainty and variation in accessibility metrics for public transport sketch planning. *Journal of Transport and Land Use*, 11(1). https://doi.org/10.5198/jtlu.2018.1074
- Curtis, T., Merritt, M., Chen, C., Perlmutter, D., Berez, D., & Ellis, B. (2019). Partnerships between transit agencies and transportation network companies (TNCs). *Transportation Research Board.* https://doi.org/10.17226/25576
- Delling, D., Pajor, T., & Werneck, R. F. (2015). Round-based public transit routing. Transportation Science, 49(3), 591–604. https://doi.org/10.1287/trsc.2014.0534
- Diab, E., Kasraian, D., Miller, E. J., & Shalaby, A. (2020). The rise and fall of transit ridership across Canada: Understanding the determinants. *Transport Policy*, 96, 101–112. https://doi.org/10.1016/j.tranpol.2020.07.002
- Diao, M., Kong, H., & Zhao, J. (2021). Impacts of transportation network companies on urban mobility. *Nature Sustainability*, 4(6), 494–500. https://doi.org/10.1038/ s41893-020-00678-z
- Dudley, G., Banister, D., & Schwanen, T. (2017). The rise of uber and regulating the disruptive innovator. *The Political Quarterly*, 88(3), 492–499. https://doi.org/ 10.1111/1467-923X.12373
- El-Geneidy, A., Levinson, D., Diab, E., Boisjoly, G., Verbich, D., & Loong, C. (2016). The cost of equity: Assessing transit accessibility and social disparity using total travel cost. *Transportation Research Part A: Policy and Practice*, 91, 302–316. https://doi. org/10.1016/j.tra.2016.07.003
- Erhardt, G. D., Roy, S., Cooper, D., Sana, B., Chen, M., & Castiglione, J. (2019). Do transportation network companies decrease or increase congestion? *Science*. *Advances*, 5(5), eaau2670. https://doi.org/10.1126/sciadv.aau2670
- Erhardt, G. D., Hoque, J. M., Goyal, V., Berrebi, S., Brakewood, C., & Watkins, K. E. (2022). Why has public transit ridership declined in the United States? *Transportation Research Part A: Policy and Practice, 161*, 68–87. https://doi.org/ 10.1016/j.tra.2022.04.006
- Farr, T. G., Rosen, P. A., Caro, E., Crippen, R., Duren, R., Hensley, S., ... Alsdorf, D. (2007). The shuttle radar topography Mission. *Reviews of Geophysics*, 45(2), RG2004. https://doi.org/10.1029/2005RG000183

Geurs, K., & van Wee, B. (2004). Accessibility evaluation of land-use and transport strategies: Review and research directions. *Journal of Transport Geography*, 12(2), 127–140. https://doi.org/10.1016/j.jtrangeo.2003.10.005

- Geurs, K., Zondag, B., de Jong, G., & de Bok, M. (2010). Accessibility appraisal of landuse/transport policy strategies: More than just adding up travel-time savings. *Transportation Research Part D: Transport and Environment*, 15(7), 382–393. https:// doi.org/10.1016/j.trd.2010.04.006
- Goodwin, P. B. (1974). Generalised time and the problem of equity in transport studies. *Transportation*, 3, 1–23. https://doi.org/10.1007/BF02351839
- Guzman, L. A., & Oviedo, D. (2018). Accessibility, affordability and equity: Assessing 'pro-poor' public transport subsidies in Bogota. *Transport Policy*, 68, 37–51. https:// doi.org/10.1016/j.tranpol.2018.04.012
- Haddad, E. A., Vieira, R. S., Jacob, M. S., Guerrini, A. W., Germani, E., Barreto, F., ... Sayon, P. L. (2019). A socioeconomic analysis of ride-hailing emergence and expansion in São Paulo. Brazil. Transportation Research Interdisciplinary Perspectives, 1. Article 100016. https://doi.org/10.1016/j.trin.2019.100016
- I, Article 100016. https://doi.org/10.1016/j.trip.2019.100016
 Hall, J. D., Palsson, C., & Price, J. (2018). Is Uber a substitute or complement for public transit? Journal of Urban Economics, 108, 36–50. https://doi.org/10.1016/j. iue.2018.09.003
- Herszenhut, D., Pereira, R. H. M., Portugal, L.d. S., Oliveira, & de Su, M. H. (2022). The impact of transit monetary costs on transport inequality. *Journal of Transport Geography*, 99, Article 103309. https://doi.org/10.1016/j.jtrangeo.2022.103309
- Hughes, R., & MacKenzie, D. (2016). Transportation network company wait times in Greater Seattle, and relationship to socioeconomic indicators. *Journal of Transport Geography*, 56, 36–44. https://doi.org/10.1016/j.jtrangeo.2016.08.014
 WCF (Ed. (2011). Singred de come demografica. 2010 (BCE)
- IBGE (Ed.). (2011). Sinopse do censo demográfico, 2010 (IBGE).
- Insardi, A., Lorenzo, R. O., Insardi, A., & Lorenzo, R. O. (2019). Measuring accessibility: A big data perspective on Uber service waiting times. *Revista de Administração de Empresas*, 59(6), 402–414. https://doi.org/10.1590/s0034-759020190606
- Jiao, J., & Wang, F. (2020). Shared mobility and transit-dependent population: A new equity opportunity or issue? *International Journal of Sustainable Transportation*, 15(4), 294–305. https://www.tandfonline.com/doi/abs/10.1080/15568318.2020.17475 78.
- Jin, S. T., Kong, H., Wu, R., & Sui, D. Z. (2018). Ridesourcing, the sharing economy, and the future of cities. *Cities*, 76, 96–104. https://doi.org/10.1016/j.cities.2018.01.012
- Jin, S. T., Kong, H., & Sui, D. Z. (2019). Uber, public transit, and urban transportation equity: A case study in new York City. *The Professional Geographer*, 71(2), 315–330. https://doi.org/10.1080/00330124.2018.1531038
- Kirk, D. S., Cavalli, N., & Brazil, N. (2020). The implications of ridehailing for risky driving and road accident injuries and fatalities. *Social Science & Medicine*, 250, Article 112793. https://doi.org/10.1016/j.socscimed.2020.112793
- Kong, H., Zhang, X., & Zhao, J. (2020). How does ridesourcing substitute for public transit? A geospatial perspective in Chengdu. *China. Journal of Transport Geography*, 86, Article 102769. https://doi.org/10.1016/j.jtrangeo.2020.102769

Lavieri, P. S., & Bhat, C. R. (2019). Investigating objective and subjective factors influencing the adoption, frequency, and characteristics of ride-hailing trips. *Transportation Research Part C: Emerging Technologies*, 105, 100–125. https://doi.org/ 10.1016/j.trc.2019.05.037

- Li, W., Shalaby, A., & Habib, K. N. (2021). Exploring the correlation between ride-hailing and multimodal transit ridership in Toronto. *Transportation*. https://doi.org/ 10.1007/s11116-021-10193-5
- Liao, Y. (2021). Ride-sourcing compared to its public-transit alternative using big trip data. Journal of Transport Geography, 95, Article 103135. https://doi.org/10.1016/j. jtrangeo.2021.103135
- Liu, D., & Kwan, M.-P. (2020). Measuring job accessibility through integrating travel time, transit fare and income: A study of the Chicago Metropolitan Area. *Tijdschrift voor Economische en Sociale Geografie*, 111(4), 671–685. https://doi.org/10.1111/ tesg.12415
- Lo, H. K., Yip, C. W., & Wan, K. H. (2003). Modeling transfer and non-linear fare structure in multi-modal network. *Transportation Research Part B: Methodological*, 37 (2), 149–170. https://doi.org/10.1016/S0191-2615(02)00005-X

Lucas, K., Van Wee, B., & Maat, K. (2016). A method to evaluate equitable accessibility: Combining ethical theories and accessibility-based approaches. *Transportation*, 43, 473–490.

Martens, K., & Di Ciommo, F. (2017). Travel time savings, accessibility gains and equity effects in cost–benefit analysis. *Transport Reviews*, 37(2), 152–169. https://doi.org/ 10.1080/01441647.2016.1276642

Motte-Baumvol, B., Aguilera, A., Bonin, O., & Nassi, C. D. (2016). Commuting patterns in the metropolitan region of Rio de Janeiro. What differences between formal and informal jobs? *Journal of Transport Geography*, 51, 59–69. https://doi.org/10.1016/j. jtrangeo.2015.10.019

- Onono, M. A., Wahome, S., Wekesa, P., Adhu, C. K., Waguma, L. W., Serem, T., ... Ong'wen, P. (2019). Effects of an expanded Uber-like transport system on access to and use of maternal and newborn health services: Findings of a prospective cohort study in Homa Bay. *Kenya*. *BMJ Global Health*, 4(3), Article e001254. https://doi. org/10.1136/bmigh-2018-001254
- Palm, M., Farber, S., Shalaby, A., & Young, M. (2021). Equity analysis and new mobility technologies: Toward meaningful interventions. *Journal of Planning Literature*, 36(1), 31–45. https://doi.org/10.1177/0885412220955197
- Pereira, R. H. M. (2019). Future accessibility impacts of transport policy scenarios: Equity and sensitivity to travel time thresholds for Bus Rapid Transit expansion in Rio de Janeiro. Journal of Transport Geography, 74, 321–332. https://doi.org/ 10.1016/j.jtrangeo.2018.12.005

Pereira, R. H., Schwanen, T., & Banister, D. (2017). Distributive justice and equity in transportation. *Transport Reviews*, 37(2), 170–191.

- Pereira, R. H. M., Saraiva, M., Herszenhut, D., Braga, C. K. V., & Conway, M. W. (2021). r5r: Rapid realistic routing on multimodal transport networks with R⁵ in R. Transport Findings, 21262. https://doi.org/10.32866/001c.21262
- Qiao, S., Zhang, M., & Yeh, A. G. O. (2023). Mind the gender gap in ride-hailing from the demand side. *Journal of Transport Geography*, 107, Article 103531.
- Rayle, L., Dai, D., Chan, N., Cervero, R., & Shaheen, S. (2016). Just a better taxi? A survey-based comparison of taxis, transit, and ridesourcing services in San Francisco. *Transport Policy*, 45, 168–178. https://doi.org/10.1016/j.tranpol.2015.10.004
- Transport Policy, 45, 168–178. https://doi.org/10.1016/j.tranpol.2015.10.004
 Sabogal-Cardona, O., Oviedo, D., Scholl, L., Crotte, A., & Bedoya-Maya, F. (2021). Not
 my usual trip: Ride-hailing characterization in Mexico City. *Travel Behaviour and
 Society*, 25, 233–245.
- Schwieterman, J. P. (2019). Uber economics: evaluating the monetary and travel time trade-offs of transportation network companies and transit service in Chicago. *Illinois. Transportation Research Record*, 2673(4), 295–304. https://doi.org/10.1177/ 0361198119839344

Schwieterman, J. P., & Livingston, M. (2019). A Review of Partnerships between Transportation Network Companies and Public Agencies in the United States (No. 19–04647). Article 19–04647. Transportation Research Board 98th Annual MeetingTransportation Research Board. https://trid.trb.org/view/1572575.

- Shaheen, S., & Chan, N. (2016). Mobility and the sharing economy: Potential to facilitate the first- and last-mile public transit connections. *Built Environment*, 42(4), 573–588. https://doi.org/10.2148/benv.42.4.573
- Shaheen, S., & Cohen, A. (2019). Shared ride services in North America: Definitions, impacts, and the future of pooling. *Transport Reviews*, 39(4), 427–442. https://doi. org/10.1080/01441647.2018.1497728

- Shokoohyar, S., Sobhani, A., & Nargesi, S. R. R. (2020). On the determinants of Uber accessibility and its spatial distribution: Evidence from Uber in Philadelphia. WIREs Data Mining and Knowledge Discovery, 10(4), Article e1362. https://doi.org/10.1002/ widm.1362
- Souza, C.d. O., Vitorino Guimarães, G., da Cruz Saldanha, L. E., Almeida Corrêa do Nascimento, F., Floriano dos Santos, T., & Vieira da Silva, M. A. (2021). Analysis of job accessibility promoted by ride hailing services: A proposed method. *Journal of Transport Geography*, 93, Article 103048. https://doi.org/10.1016/j. itrangeo.2021.103048
- Sui, Y., Zhang, H., Song, X., Shao, F., Yu, X., Shibasaki, R., Sun, R., Yuan, M., Wang, C., Li, S., & Li, Y. (2019). GPS data in urban online ride-hailing: A comparative analysis on fuel consumption and emissions. *Journal of Cleaner Production*, 227, 495–505. https://doi.org/10.1016/j.jclepro.2019.04.159
- Sunitiyoso, Y., Rahayu, W. A., Nuraeni, S., Nurdayat, I. F., Pambudi, N. F., & Hadiansyah, F. (2022). Role of ride-hailing in multimodal commuting. *Case Studies* on Transport Policy.. https://doi.org/10.1016/j.cstp.2022.04.013
- Tirachini, A. (2020). Ride-hailing, travel behaviour and sustainable mobility: An international review. *Transportation*, 47(4), 2011–2047. https://doi.org/10.1007/ s11116-019-10070-2
- Tirachini, A., & del Río, M. (2019). Ride-hailing in Santiago de Chile: Users' characterisation and effects on travel behaviour. *Transport Policy*, 82, 46–57. https:// doi.org/10.1016/j.tranpol.2019.07.008
- Tirachini, A., & Gomez-Lobo, A. (2020). Does ride-hailing increase or decrease vehicle kilometers traveled (VKT)? A simulation approach for Santiago de Chile. *International Journal of Sustainable Transportation*, 14(3), 187–204. https://doi.org/ 10.1080/15568318.2018.1539146
- Wang, M., & Mu, L. (2018). Spatial disparities of Uber accessibility: An exploratory analysis in Atlanta, USA. Computers, Environment and Urban Systems, 67, 169–175. https://doi.org/10.1016/j.compenvurbsys.2017.09.003
- Wang, M., Martin, E. W., & Shaheen, S. A. (2012). Carsharing in Shanghai, China: Analysis of behavioral response to local survey and potential competition. *Transportation Research Record*, 2319(1), 86–95. https://doi.org/10.3141/2319-10
- Warwar, L., & Pereira, R. H. M. (2022). Tendências e desigualdades da mobilidade urbana no Brasil II: Características e padrões de consumo da mobilidade por aplicativo. http://www.ipea.gov.br.https://repositorio.ipea.gov.br/handle/11 058/11240.
- Yan, X., Levine, J., & Zhao, X. (2019). Integrating ridesourcing services with public transit: An evaluation of traveler responses combining revealed and stated preference data. *Transportation Research Part C: Emerging Technologies*, 105, 683–696. https://doi.org/10.1016/j.trc.2018.07.029
- Young, M., & Farber, S. (2019). The who, why, and when of Uber and other ride-hailing trips: An examination of a large sample household travel survey. *Transportation Research Part A: Policy and Practice*, 119, 383–392. https://doi.org/10.1016/j. tra.2018.11.018
- Young, M., & Farber, S. (2020). Using wait-time thresholds to improve mobility: The case of UberWAV Services in Toronto. *Findings*. https://doi.org/10.32866/001c.14547
- Young, M., & Farber, S. (2021). Ride-hailing platforms are shaping the future of mobility, but for whom? The platform economy and the City: Urban peril and promise in the new digital economy edited by Austin Zwick and Zachary Spicer. McGill-Queens University Press.
- Young, M., Allen, J., & Farber, S. (2020). Measuring when Uber behaves as a substitute or supplement to transit: An examination of travel-time differences in Toronto. *Journal* of *Transport Geography*, 82, Article 102629. https://doi.org/10.1016/j. itrangeo.2019.102629
- Yu, B., Ma, Y., Xue, M., Tang, B., Wang, B., Yan, J., & Wei, Y.-M. (2017). Environmental benefits from ridesharing: A case of Beijing. *Applied Energy*, 191, 141–152. https:// doi.org/10.1016/j.apenergy.2017.01.052
- Zhang, M., & Zhao, P. (2021). Literature review on urban transport equity in transitional China: From empirical studies to universal knowledge. *Journal of Transport Geography*, 96, Article 103177.
- Zuniga-Garcia, N., Gurumurthy, K. M., Yahia, C. N., Kockelman, K. M., & Machemehl, R. B. (2022). Integrating shared mobility services with public transit in areas of low demand. *Journal of Public Transportation*, 24, Article 100032. https:// doi.org/10.1016/j.jpubtr.2022.100032